Author’s response to reviews

Title: Intramural haematoma of the thoracic aorta: who is to be alerted the cardiologist or the cardiac surgeon?

Authors:

NIKOLAOS G. BAIKOSSIS (ngbaik@yahoo.com)
EFSTRATIOS E. APOSTOLAKIS (stratisapostolakis@yahoo.gr)
STAVROS N. SIMINELAKIS (ssiminel@yahoo.gr)
GEORGIOS S. PAPADOPOULOS (baikoussisn@yahoo.com)
JOHN GOUDEVENOS (ngbaik@yahoo.com)

Version: 4 Date: 18 September 2009

Author’s response to reviews:

Dear Editor

We would like to inform you that we correct the paper according your reccomandations. We added the references 17 and 43 correctly, we wrote the competing interes and the contributions, we corrected the authors name.